Author's response to reviews

Title: Perspectives of staff nurses about the reasons for and nature of patient-initiated call-lights: An exploratory survey study in four U.S. hospitals

Authors:

Huey-Ming Tzeng (tzeng_hueyming@yahoo.com)

Version: 2 Date: 18 August 2009

Author's response to reviews: see over
2009/08/18

Dear Dr. Melissa Norton, Editor-in-Chief:

I am writing to submit my revised paper to *BMC Health Services Research* for publication evaluation. The title of the paper is “Perspectives of staff nurses about the reasons for and nature of patient-initiated call-lights: An exploratory survey study in four U.S. hospitals.”

Based on the e-mail from Dr. Iratxe Puebla (dated August 18, 2009), I have made the changes on the abstract and add a new section of Competing Interests. All changes are marked in RED for your easy reference.

This manuscript is submitted solely to *BMC Health Services Research*, and the work has not been published elsewhere.

Thank you in advance for your consideration. Looking forward to hearing from you.

Sincerely,

Author